Author's response to reviews

Title: Health care reform and job satisfaction of primary health care physicians in Lithuania

Authors:

Aurelija AB Blazeviciene (blazeviciene@yahoo.com)
Ilona IB Buciuniene (ilona.buciuniene@ism.lt)
Egle EK Kaubriene (eglutebl@yahoo.com)

Version: 5 Date: 25 February 2005

Author's response to reviews: see over
Dear editor,

Thank you for comments about our article. It was really very helpful to get your opinion.

We accomplished next correction: changed citation in the text and references list. So are we paste the tables in the main manuscript text after the references.

Respectfully yours,

Aurelija Blaževičienė